Author's response to reviews

Title: Assessment of nerve involvement in the lumbar spine: association between magnetic resonance imaging, physical examination and pain drawing findings

Authors:

Bo C Bertilson (bo.bertilson@ki.se)
Eva Brosjö (evaxray@hotmail.com)
Hans Billing (hb@telemedicineclinic.com)
Lars-Erik Strender (Lars-Erik.Strender@ki.se)

Version: 7 Date: 30 July 2010

Author's response to reviews:

We have tried to make all changes according to your instructions concerning, the abstract, figures, tables and additional files as well as concerning line numbers and typography and spelling. However we have not been able to use the Tables tool you mention to take away the shaded areas. The shaded areas are shaded simply to indicate that there is no data available in these columns and rows. Could you please help us on this matter and suggest a solution.